Author's response to reviews

Title: Weight loss intervention for individuals with high internal disinhibition: Design of the Acceptance Based Behavioral Intervention (ABBI) randomized controlled trial.

Authors:

Jason Lillis (jasonlillis22@gmail.com)
Heather M Niemeier (niemeieh@uww.edu)
Kathryn M Ross (kross3@lifespan.org)
John G Thomas (jthomas4@lifespan.org)
Tricia Leahey (tricia.leahey@unconn.edu)
Jessica Unick (junick@lifespan.org)
Kathleen E Kendra (kkendra10@gmail.com)
Rena R Wing (rwing@lifespan.org)

Version: 2 Date: 26 March 2015

Author's response to reviews:

All requested revisions are addressed in the cover letter. Changes in the manuscript are: (1) the addition of a clinicaltrials.gov registration date, (2) the addition of a "competing interests" section to the manuscript, (3) the addition of a "author’s contributions" section to the manuscript.